# Geochemistry, Geophysics, Geosystems



ŗ

## **RESEARCH ARTICLE**

10.1029/2019GC008837

#### **Key Points:**

- We summarize available geophysical and petrological proxies for the thermal structure of southern Tibet
- Simple two-dimensional thermal models are capable of matching the majority of available constraints
- Average convergence across the Himalayas must have been faster than the present-day rate

**Supporting Information:** 

Supporting Information S1

#### Correspondence to:

T. J. Craig, t.j.craig@leeds.ac.uk

#### **Citation**:

Craig, T. J., Kelemen, P. B., Hacker, B. R., & Copley, A. (2020). Reconciling geophysical and petrological estimates of the thermal structure of southern Tibet. *Geochemistry, Geophysics, Geosystems, 21*, e2019GC008837. https://doi.org/10.1029/2019GC008837

Received 25 NOV 2019 Accepted 2 JUL 2020 Accepted article online 6 JUL 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Reconciling Geophysical and Petrological Estimates of the Thermal Structure of Southern Tibet

## T. J. Craig<sup>1</sup>, P. B. Kelemen<sup>2</sup>, B. R. Hacker<sup>3</sup>, and A. Copley<sup>4</sup>

<sup>1</sup>COMET, Institute of Geophysics and Tectonics, School of Earth and Environment, University of Leeds, Leeds, UK, <sup>2</sup>Lamont-Doherty Earth Observatory, Columbia University, Palisades, NY, USA, <sup>3</sup>Department of Earth Science, University of California, Santa Barbara, CA, USA, <sup>4</sup>COMET, Department of Earth Sciences, University of Cambridge, Cambridge, UK

Abstract The thermal structure of the Tibetan plateau—the largest orogenic system on Earth—remains largely unknown. Numerous avenues provide fragmentary pressure/temperature information, both at the present (predominantly informed though geophysical observation) and on the evolution of the thermal structure over the recent past (combining petrological, geochemical, and geophysical observables). However, these individual constraints have proven hard to reconcile with each other. Here, we show that models for the simple underthrusting of India beneath southern Tibet are capable of matching all available constraints on its thermal structure, both at the present day and since the Miocene. Many parameters in such models remain poorly constrained, and we explore the various trade-offs among the competing influences these parameters may have. However, three consistent features to such models emerge: (i) that present-day geophysical observations require the presence of relatively cold underthrust Indian lithosphere beneath southern Tibet; (ii) that geochemical constraints require the removal of Indian mantle from beneath southern Tibet at some point during the early Miocene, although the mechanism of this removal, and whether it includes the removal of any crustal material, is not constrained by our models; and (iii) that the combination of the southern extent of Miocene mantle-derived magmatism and the present-day geophysical structure and earthquake distribution of southern Tibet require that the time-averaged rate of underthrusting of India relative to central Tibet since the middle Miocene has been faster than it is at present.

**Plain Language Summary** Numerous proxies for the temperature structure of the lithosphere beneath southern Tibet exist. Geophysical observations provide information about the present-day structure, geometry, and thermal structure of the area; geodetic and geological observations constrain the regional kinematics over the recent past; geochemical and petrological observations from magmatic rocks within the Himalaya-Karakorum-Tibetan plateau provide sparse information about the temperatures beneath southern Tibet over the history of the collision zone. In this paper, we summarize the range of these observations that are presently available and present a series of thermal models aimed at exploring what parameters for the thermal evolution of southern Tibet are capable of explaining all of these observations within a coherent framework. While present-day observations require the presence of a cold India-derived lithosphere underthrust beneath southern Tibet, geochemical and petrological data require that this was not present during the early Miocene and was removed prior to this. Further, to fit all observations, our models require that the average rate of underthrusting since the middle Miocene must have been greater than that observed at the present.

#### 1. Introduction

As the largest modern continental orogeny, the Tibetan plateau represents an ideal modern analog for major collisional orogenic systems since the onset of modern-style tectonics. The temperature structure and evolution of the Tibetan plateau is of vital importance in understanding not only the seismicity, volcanism, rheology, and deformation of the modern collision zone but also the evolution of orogenic systems in general. Significant areas of the continental interiors are dominated by the geological products of such collision zones, with their chemical and rheological history governed by their thermal evolution, and the resultant metamorphic and igneous processes they were subjected to, during orogenic growth and decay. Despite this importance, the thermal structure of the lower crust and upper mantle of Tibet remains a controversial topic. Thermochronology of rocks exposed along the Himalayan front offers a window into the kinematics and shallow thermal structure of the orogenic front and Himalayan prism (e.g., Bollinger et al., 2006; Godin et al., 2001; Herman et al., 2010). However, the incorporation of material into the Himalayas only represents a relatively small volume of the total lithosphere involved in the collision zone. Constraining the thermal structure of the deeper Tibetan plateau is more difficult, and despite extensive work, there are relatively few available constraints on either its contemporary structure or its evolution through time. Many of the constraints that do exist are hard to reconcile into a consistent thermal structure, particularly at depth beneath the plateau. Similarly, and in part due to the paucity of observational constraints on the deeper rheological structure of Tibet, debate continues about the geodynamic evolution of the plateau. Processes including large-scale underthrusting, crustal channel flow, gravitational spreading, convective delamination and slab tearing, break-off, and removal are all invoked to explain particular geological, geochemical, and geophysical observations.

Here, we focus on the deep thermal evolution of the southern half of the Tibetan plateau, limited to within  $\sim$ 500 km of the Himalayan Front, hereafter referred to as "south/southern Tibet." We summarize existing proxies for the thermal and rheological structure of southern Tibet and present a series of two-dimensional models for the thermal evolution of the region from the Miocene to present, to investigate what models are compatible with available geophysical and geochemical constraints. As we shall discuss, these models do not provide a unique solution for the structure and evolution of southern Tibet but instead demonstrate that reconciling available geophysical and geochemical constraints is possible within a coherent framework and highlight some consistent features that are common to all successful models.

## 2. Geophysical Constraints

Receiver functions, when combined with co-located surface-wave dispersion studies, provide vital information on the present-day crustal geometry of the collision zone (Figures 1 and 2). In general, the crustal thickness varies from 35–40 km across much of peninsular India to 75–80 km under southern Tibet, with this increase being roughly linear over the first 200–250 km of the collision zone, north of the Himalayan front, where incoming Indian material is underthrust beneath Tibet. While there is significant local variability in this trend (e.g., Subedi et al., 2018), particularly in the region of most rapid change (and most variable geometry) beneath the high Himalaya (e.g., Hubbard et al., 2016; Ingleby et al., 2020), the available information supports a broadly two-dimensional crustal-scale pattern in southern Tibet, perpendicular to the southern edge of the Plateau.

This simple picture, of Indian lithosphere thrusting northwards beneath the Tibetan plateau in a manner that, in terms of the overall crustal thickness, varies little along strike, is also consistent with tomographic imaging of the deeper seismological structure of the plateau. Surface-wave tomography at frequencies appropriate to both crustal (Gilligan & Priestley, 2018) and mantle (M. Chen et al., 2017) levels shows a gradually thickening crustal pile beneath the Himalayas with low seismic wave speeds, an approximately uniform crustal thickness beneath the Tibetan plateau itself, and a seismically fast uppermost mantle, consistent with the presence of cold India-derived lithosphere beneath the southern plateau. Due to its different sensitivity, body-wave tomography (e.g., Li & Song, 2018; Liang et al., 2016; Ren & Shen, 2008) shows wider lateral variability in this seismological picture. Such studies commonly detect intraplateau features, such as the graben systems of southern Tibet, but match the broad image of a seismologically fast region of Indian lithosphere underlying the thickened crust of southern Tibet.

The underthrusting of Indian crustal material, along with progressive mineralogical and rheological evolution northwards, is also suggested by modeling the variation in terrestrial gravity observations across the Himalayas and onto southern Tibet (Cattin et al., 2001; Hétenyi et al., 2007). While such models only provide a relatively coarse image of the density distribution at depth beneath the plateau, and their petrological implications are subject to the poorly known thermal structure, they indicate a progressive densification of Indian crustal material over the first  $\sim$ 400 km of the plateau, most probably driven by fluid release during dehydration of underthrust Indian material as it heats up.

We now consider geophysical constraints on the temperature distribution of India and southern Tibet. With the well-established dependence of seismic slip on the temperature and composition of lithospheric material, the distribution of earthquakes within the collision zone provides an initial constraint on the thermal



## Geochemistry, Geophysics, Geosystems



**Figure 1.** (a) Earthquake depths across the India-Asia collision zone. Data compilation from Craig et al. (2012), updated with Sippl et al. (2013), Mitra et al. (2014), Paul et al. (2015), Reynolds et al. (2015), Kufner et al. (2016), Mendoza et al. (2016), Ainscoe et al. (2017), Diehl et al. (2017), Gibbons and Kvaerna (2017), Huang et al. (2017), Negi et al. (2017), Kanna et al. (2018), and Parija et al. (2018). Dashed lines highlight areas of deep seismicity associated with the Hindu-Kush deep seismic zone and the Indo-Burman subduction zone and also the northern boundary of the Tibetan plateau. (b) Receiver-function observations of crustal thickness (after Gilligan & Priestley, 2018). (c) Shear wave splitting measurements (after Agius & Lebedev, 2017). (d) Volcanism across Tibet, where both chemistry and age are well constrained (after Yakovlev et al., 2019). (e) Locations of xenolith suites (Chan et al., 2009; Ding et al., 2007; Hacker et al., 2000; Liu et al., 2011), seismological observations of the inferred ABQT (Mechie et al., 2004; Sheehan et al., 2014), and the region of low  $S_n$  propagation in northern Tibet (McNamara & Owens, 1995). Dashed white line indicates ~500 km from the Himalayan front along the India-Asia convergence vector.

structure of the plateau (Figures 1a and 2b). Across much of the Tibetan plateau itself, the depth extent of small- and moderate-magnitude earthquakes (Craig et al., 2012; Langin et al., 2003; Molnar & Chen, 1983; Sloan et al., 2011), the rupture extent of large earthquakes (Elliott et al., 2010; Wright et al., 2013), and the geodetically determined locking depth of major crustal faults (Wright et al., 2013) all terminate at ~12- to 13-km depths, with this transition in crustal rocks typically occurring at ~350°C (W.-P. Chen & Molnar, 1983; Jackson et al., 2008).

In peninsular India, seismicity is observed to occur throughout the thickness of the crust in all areas where major seismic activity has been observed (northwestern peninsular India, around the Shillong plateau, and beneath the Himalayan foreland) and in some cases into the top of the uppermost lithospheric mantle (Craig et al., 2012; Monsalve et al., 2006; Schulte-Pelkum et al., 2019). Where Indian material is underthrust beneath the Tibetan plateau, a narrowing tongue of seismicity extends beneath the Himalayas and southern Tibet following the Indian lower crust and Moho (Monsalve et al., 2006; Schulte-Pelkum et al., 2019), gradually dying out at 400–500 km north of the Himalayan front (Craig et al., 2012; Priestley et al., 2008). These





**Figure 2.** Cross sections though southern Tibet, along the line shown in Figure 1a. All horizontal distances are given as distance from the local Himalayan front along the present-day convergence vector (shown on Figure 1), and therefore correct for the curvature fo the Himalayan front. (a) Topography along a 20-km-wide swath. (b) Earthquake depths and receiver-function-based crustal thickness estimates. (c) Shear-wave splitting measurements, showing both amplitude and azimuth. Null results are shown by open circles along the *x*-axis. (d) Proxies for the thermal structure of southern Tibet. All data are plotted at their present location, uncorrected for their sampling time. For visibility, volcanism values with Mg# > 0.7 are plotted above the line, and those with Mg# < 0.7 below. (e) Xenolith data for Namring (Chan et al., 2009). Dashed boxes show P/T uncertainties. (f) Xenolith data for Sailipu (Liu et al., 2011). Inset histogram shows the temperature population of the xenolith suite. (g) Xenolith data for Yibuchaka (Ding et al., 2007). Inset histogram shows the temperature population of the xenolith suite. (g) Xenolith data for Taipinghu (Hacker et al., 2000). Red line is the temperature profile used in thermal models for the northern boundary condition. Inset dates on panels (e)–(h) show the age of emplacement, with the age of peak metamorphism, if known, given in brackets. See Figure 1 for the locations of xenolith suites shown in panels (e)–(h).

earthquakes represent the internal deformation of underthrust Indian lithosphere. The rate of earthquake occurrence is low and is concentrated into two regional groups in northwestern and eastern South Tibet. This patchy distribution may either reflect the slow strain rates in the host material and hence may not be a representative seismic catalog (similar to that seen in peninsular India) or a further control on seismicity based on spatial variations in the mineralogical structure of underthrust India (Alvizuri & Hetényi, 2019; Jamtveit et al., 2019). In general, seismicity associated with either the upper mantle (Boettcher et al., 2007; Craig et al., 2014) or a mafic and anhydrous lower crust (Jackson et al., 2008; Mackwell, 1998) can persist to temperatures of at least 600–650°C, before, at standard tectonic strain rates, deformation becomes ductile. Beneath the Himalaya, the pattern in seismicity is more complex, as shown on Figure 2, accommodating this bifurcation into shallow upper crustal seismicity, and deeper seismicity roughly along the Moho

Satellite magnetic measurements (Alsdorf & Nelson, 1999), although lacking in spatial resolution, show a magnetic low across the plateau associated with a shallow depth ( $\sim$ 15 km) of the crustal Curie temperature ( $\sim$ 550°C at 15 km, from which they infer that temperatures of 600–650°C are reached regionally at 16- to 18-km depth; Figure 2d).

Limited surface heat flow measurements from across Tibet offer a direct estimate of the shallow geotherm (International Heat Flow Commission, 2011). However, heat flow measurements are logistically complex and expensive, are often taken in nonrepresentative sections of the plateau (rift basins, geothermal

areas, etc.), and may be strongly influenced by shallow radiogenic structure or recent magmatism. These factors result in a wide distribution of values, spanning a range of 200 mW m<sup>-2</sup>. As a result of this uncertainty over which values may be representative of the deeper thermal structure of the plateau, we exclude surface heat flow measurements from within the plateau from the rest of our analysis.

Seismological data provide a range of constraints on the present-day temperature structure of the plateau (Figure 2d). Densely spaced seismic acquisitions in the northern Himalayas (HIMNT; Sheehan et al., 2014) and southern Tibet (InDEPTH III; Mechie et al., 2004) detected high reflectivity contrasts in  $V_p/V_s$  ratios linked to high seismic attenuation, inferred to represent the transition from  $\alpha$  to  $\beta$  quartz within the crust (hereafter referred to as the ABQT). While questions remain about why this transition has not been detected more commonly across the plateau, laboratory constraints on the pressure-temperature controls of this transition (Angel et al., 2017) allow an estimate of the temperature at the depths where this interface is inferred, at depths from ~45–20 km, shallowing northwards (Figures 1e and 2d).

The variation in seismic anisotropy revealed by both shear-wave tomography and shear-wave splitting across the plateau provides critical information on the geometry of strain. Recent shear-wave tomography (Gilligan & Priestley, 2018) shows the development of significant radial anisotropy in the midcrust of the central Tibetan plateau, in contrast to the seismic structure of both India and southernmost Tibet, where anisotropy is low. Similarly, shear-wave splitting (Figures 1c and 2c; after Agius & Lebedev, 2017), while showing wide variation across the collision zone (which may partly reflect the range of data selection and analysis methods used), shows a consistent northwards increase in amplitude and, more importantly, a shift toward a consistent preferential alignment in the fast direction, such that more than  $\sim$  300 km north of the Himalayan front, the fast direction is consistently 070-090°E. The development of a coherent anisotropic structure within the central plateau is suggestive of a process in which gradual heating of underthrust Indian lithosphere, as it moves northwards beneath Tibet, leads to its progressive rheological weakening and softening. This weakening allows increasing volumes of crustal and mantle material (starting with the midcrust) to deform and flow, resulting in a situation where a weaker, ductile material from central and northern Tibet flows out of the collision zone to the east, leading to the development of a preferred mineral orientation, resulting in both a coherent shear-wave splitting fabric and midcrustal anisotropy (e.g., Gilligan & Priestley, 2018; Pandey et al., 2015).

Beneath northern Tibet, frequency-dependent inefficient propagation of the seismic phases  $P_n$  and  $S_n$  (head waves that propagate along the Moho) indicates an inverted gradient in seismic velocities beneath the Moho, over a depth extent that increases northwards (Figure 2d; Barron & Priestley, 2009; McNamara & Owens, 1995; Ni & Barazangi, 1983). While this observation does not provide an absolute estimate of the temperature at this point, it does suggest an inversion of the vertical temperature gradient across the Moho (ie., the upper mantle just below the Moho may be hotter than the mantle beneath it).

## 3. Petrological and Geochemical Constraints

While geophysical observations are limited to direct constraints on the present-day structure of the plateau, geochemical information offers the opportunity to constrain the past temperature at depth within the plateau. In all cases, the present-day surface location of the samples providing these geochemical constraints is displaced relative to the deep structure of the plateau at the time they were emplaced, due to subsequent motion within the collision zone. Hence, sample locations must be corrected for the movement of material within the collision zone in the intervening time since their age of chemical equilibration.

Southern Tibet has experienced numerous phases of volcanism over the last 50 Myr. Here, we draw on a recent compilation by Yakovlev et al. (2019) for volcanic rocks within the Tibetan plateau for which both detailed geochemical and chronological data are available (Figures 1d and 2d). Erupted volcanics with Mg#  $\ll$  0.7, particularly those in northern Tibet, likely reflect either intracrustal melting (for more recent samples), the effects of arc volcanism (in older samples), or highly fractionated mantle-derived melts. While intracrustal melting has the potential to add some further constraints on the recent temperature at depth within the plateau (particularly at midcrustal depths), this would require a level of detailed analysis (and knowledge of the source rock composition) that is beyond the scope of this study. Instead, we exclude samples with Mg# < 0.7 and focus on those samples with Mg# > 0.7 (in southern Tibet, these are typically potassic or ultrapotassic rocks), as a method for identifying samples which likely have equilibrated with

refractory peridotite (Mg# > 0.89), and hence derive from the melting of a source at subcrustal levels. The generation of such melts requires temperatures sufficient to partially melt a source rock—either mantle peridotite or various metasomatic or subducted components—and to equilibrate that melt with mantle peridotite. Obviously, such processes must take place at depths where mantle peridotite is present, below the Moho. We then assume that the duration of intracrustal storage for such rocks is negligible compared to the timespan of the India-Asia collision (e.g., Mutch et al., 2019), and thus that eruption ages are representative of the age of melting. This argument is supported by the requirement for a high Mg#, indicating that the magma has undergone limited fractionation and/or crustal assimilation. The spatial distribution of such volcanics hence constrains the age and extent of widespread (but predominantly small melt-fraction) mantle melting beneath the plateau. The detailed data set, including a breakdown of Mg#, rather than the binary division used here, is shown in the supporting information Figure S1 and fully described in Yakovlev et al. (2019).

Much of the older volcanism across the plateau prior to 50 Ma is likely associated with precollisional arc-type volcanism in the accreted terranes of central Tibet. From 50 to ~30 Ma, central and southern Tibet show little magmatic activity. After 30 Ma, there is a progressive spread of magmatism, including magmatism with Mg# > 0.7, from central Tibet both northwards and southwards. In northern Tibet, magmatic activity with Mg# < 0.7 continues to the present day. In southern and central Tibet, the majority of magmatic activity (including that with Mg# > 0.7) terminates at or before ~12 Ma (and predominantly by 15 Ma). This general trend in magmatic activity is mirrored along the length of Himalayan arc, although there is some variability in the time at which various phases of this behavior occurred, suggesting that the underlying geodynamic processes were diachronous along strike (Webb et al., 2017). Crucially, the extent of Mg# > 0.7 magmatism in south and central Tibet into the mid-Miocene indicates that the mantle beneath this region at this time was hot enough, and shallow enough, to melt—a feature that ends by 12 Ma.

More accurate constraints on temperature at depth can be obtained from detailed analysis of xenoliths contained within some of these volcanics. Here, we draw on the four published sets of xenolith data for Tibet (excluding the Pamir): from Taipinghu in the northern Qiangtang terrane (Hacker et al., 2000); Yibuchaka, southern Qiangtang terrane (Ding et al., 2007); Namring, just north of the Indus-Yarlung suture (Chan et al., 2009); and Sailipu, western Lhasa terrane (Liu et al., 2011), all shown on Figure 1e.

In the case of Sailipu (Liu et al., 2011), the xenoliths are peridotite, entrained in an ultrapotassic host rock. The ultrapotassic lava most probably formed via melting of mantle material. For Yibuchaka (Ding et al., 2007), the xenoliths are a mixture of mafic granulites, lherzolites, and websterites and therefore span the Moho, from lower crust to uppermost mantle. While accurate temperature estimates exist for samples from these two locations (see Figures 2f and 2g), the pressure constraints for the Sailipu and Yibuchaka xenolith suites are based on the absence of garnet, based on which they are inferred to originate above the garnet-spinel transition and are therefore only an upper limit. For the xenoliths from Yibuchaka and Sailipu, only eruption ages are known (28 and 17 Ma, respectively), not the age at which the xenoliths represent the thermal structure of the plateau at, or shortly before, their time of emplacement, or instead reflect prior processes, (and potentially arc-related volcanism), that affected the plateau. We therefore do not consider them further.

The best-constrained set of xenolith data for southern Tibet comes from an andesitic/potassic dyke near Namring (Chan et al., 2009). These comprise micaceous, felsic, and mafic granulite, and ultramafic xenoliths. All have well-constrained temperature estimates, and three have well-constrained pressure estimates (shown on Figure 2e). Crucially, both the age of emplacement (12.7 Ma) and the age of peak metamorphism (16.8, 15.6, and 14.4 Ma for the three separate xenoliths shown in Figure 2e) are known. Hence, the pressure/temperature estimates from this xenolith suite can be used to constrain the thermal structure of the plateau in that time range.

Recently erupted crustal (metasedimentary and mafic) xenoliths from Taipinghu (Hacker et al., 2000) with robust thermobarometry provide information on the near-present-day geotherm of north-central Tibet (Figure 2h). Additionally, the near-anhydrous nature of these xenoliths helps to explain why, despite the high inferred temperatures, the Tibetan midcrust does not host widespread melt.

Significant work has been done by numerous authors in constraining the thermochronological and thermobarometric evolution of the Himalayas, in many cases as a way of investigating the structural evolution and sequence of faulting in the Himalayan prism. Much of this has focused on low temperature (e.g.,  ${}^{40}$ Ar/ ${}^{39}$ Ar; apatite fission track, U-Th-He) thermometers and do not constrain the deeper thermal structure of the plateau. Additionally, as we shall discuss, the finer-scale detail of the faulting regime in the Himalayas, while having a major influence on the shallow thermal structure of the mountain range, has little impact on the lower crustal thermal structure of southern Tibet.

## 4. The Thermal Structure of Southern Tibet

A difficulty with southern Tibet lies in reconciling the geophysical data, that imply a relatively intact India underthrust beneath the southern plateau, with geochemical data that require widespread mantle melting beneath areas of the southern plateau until  $\sim$ 10 Ma (Chung et al., 2005; Yakovlev et al., 2019), with inferred temperatures around the Moho that are inconsistent with the underthrusting of cool Indian lithosphere.

One proposed solution has been the removal of large amounts of lithospheric mantle from beneath the plateau, most probably through the break-off of underthrust Indian lithosphere (DeCelles et al., 2002; Guillot et al., 2003; Replumaz et al., 2010). Over the last decade, increasing evidence, ranging from the identification of a relict "Indian" slab in the deeper mantle beneath Tibet (Replumaz et al., 2010, 2014), the progressive migration and termination of magmatic activity across Tibet (Chung et al., 2005; Webb et al., 2017; Yakovlev et al., 2019), changes in sedimentation in both the Himalayan foreland (Mugnier & Huyghe, 2006), in Northern India (Najman et al., 2018), and within the plateau interior (Carrapa et al., 2014; Leary et al., 2016), and a switch from contraction to extension (Stearns et al., 2013), all support a model in which an initial phase of underthrusting by colder Indian material was followed by a prolonged phase of southward slab retreat back under southern Tibet, ending with at least one break-off, which was then followed by renewed underthrusting of Indian material beneath the plateau—a model with major implications for the present-day temperature field beneath the plateau in the lower crust and upper mantle.

These observations therefore raise the question of whether a phase of slab break-off, and the implied removal of Indian material from beneath the plateau, is compatible with the presence of hot upper mantle recently enough, and shallowly enough, to match the geochemical data described above, while also being able, following re-underthrusting of continental lithosphere from India, to match geophysical constraints on the present-day plateau structure. To address this question, we model the thermal evolution of the southern Tibetan plateau, focusing on the last 20 Myr, to investigate the range of options for the kinematic evolution of the plateau that fit all available geophysical and geochemical constraints on the geological and thermal structure of the southern Tibetan plateau. Here, we specifically focus on the thermal structure at depths greater than  $\sim$ 20 km—an approach that allows us to simplify the shallow fault-dominated kinematics of the collision zone as expressed in the complexity of Himalayan tectonics.

#### 5. Thermal Modeling

We construct a set of two-dimensional forward models for the thermal evolution of the southern Tibetan plateau, solving for the advection, diffusion, and internal production of heat through time using

$$\frac{\partial \left[\rho(T) \ C_p(T) \ T\right]}{\partial t} = \nabla \cdot (k(T) \nabla T) + \mathbf{v} \cdot \nabla T + H_r,\tag{1}$$

where *T* is temperature, *t* is time, **v** is the advective velocity field,  $H_r$  is radiogenic heat production, and  $\rho$ ,  $C_p$ , and *k* are the density, heat capacity, and thermal conductivity, respectively. We use a finite difference approach and separately solve for the diffusive and advective terms at each time step using operator splitting. An iterative Crank-Nicolson approach is used for the diffusive step and an upwind scheme for the advective step.

In the following sections, we show that it is possible to fit all available constraints with a range of combinations of model parameters. We discuss the uniqueness of the parameter choices we make, the various trade-offs among parameters, and what ranges are plausible for those parameters. Our models are kinematic, rather than dynamic, and the model geometry and the advective velocity field are prescribed. We follow previous approaches (e.g., Bollinger et al., 2006; Cattin et al., 2001; Craig et al., 2012; Henry et al., 1997; Herman et al., 2010; P. Nábělek & Nábělek, 2014; Wang et al., 2013) in modeling a scenario in which Indian material is underthrust beneath Asia-derived crustal material, with the accretion of Indian upper crustal material into the Himalayan prism, and with the erosive removal of material in the Himalayas. Model kinematics and boundary conditions are summarized in Figure 3a.

For completeness, we initially run our model, with cool Indian material being underthrust beneath Tibet, for  $(50-t_b)$  Myr, where  $t_b$  is the time of material removal. As an approximation to the removal of Indian material through slab break-off (or by any other mechanism), at time  $t_b$ , we replace all Indian-derived material beneath a given depth, and north of  $x_b$  km beyond the Himalayan front, with mantle material at the temperature of the mantle adiabat (Figure 3b). We then continue underthrusting Indian material for a further  $t_b$  Myr, corresponding to the time between "break-off" and the present day, during which the emplaced material is allowed to cool conductively, until we reach the model "present day" (Figure 3c).

Major uncertainties in such models arise from a variety of underlying causes: The thermal profiles used as boundary conditions for incoming "India lithosphere" and overthrusting "Tibetan crust" the distribution of radiogenic heat production, particularly in the Tibetan overthrust; the model kinematics, their partitioning between overthrusting, underplating through the Himalayan megathrust, and the removal of material through the Himalayan front, and their evolution through time; and the timing, location, and mode of removal from the lower sections of the plateau interior. We address each of these in turn in the following sections, describing the range of plausible values, and the effects that they have on model outputs.

For the thermal structure of incoming Indian lithosphere, we use a model in one-dimensional steady state, based on cratonic areas of Indian south of the Himalayan front (Figure 4). We assume a lithospheric thickness of 200 km, consistent with geophysical estimates based on the variation of shear wave speed with depth, and with pressure-temperature estimates from a suite of mantle xenoliths from Wajrakarur, beneath the Dharwar craton (Roy & Mareschal, 2011). In all cases, we use  $0.02 \,\mu W \,m^{-3}$  for the lithospheric mantle heat production, although this value has a negligible effect. Constraints on crustal radiogenic heat production are sparse, largely informed by surface exposure of geological units believed to be representative of the deeper crustal column. In consequence, we therefore trial a range of models using crustal heat production columns based on surface measurements of radiogenic heat production, combined with surface heat flow measurements.

In our preferred model of the initial steady-state Indian thermal structure (shown in Figure 4), based on crustal data from the cratonic western Dharwar province (Roy & Mareschal, 2011), we use a 19-km upper crust of tonalite-trondhjemite-granodiorite gneiss, overlying a 19-km lower crust of mafic granulite, with radiogenic heat production of 1.1 and 0.36  $\mu$ W m<sup>-3</sup>, respectively, yielding a surface heat flux of 41.0 mW m<sup>-2</sup>. Given the paucity of comprehensive data on crustal structure and heat production, we also trial three further models (Figure 4) for crustal heat production, based on values for the eastern Dharwar (Roy & Mareschal, 2011) and Bundelkhand (Podugu et al., 2017) provinces, all summarized in Table S1. Figure 5 then shows a comparison between our preferred model (western Dharwar) and the other three models. The model for Bundelkhand (Figures 4 and 5d) is significantly colder than any for Dharwar in terms of upper mantle temperatures (Moho temperature of ~460°C, reaching 600°C at 56-km depth) and is likely too cold to represent the averaged thermal structure of India, as it would result in an extremely deep seismogenic thickness from, and an incredibly strong lithosphere-neither of which is supported by continent-wide data. Models for eastern Dharwar, with significantly higher radiogenic heat production in the upper Indian crust (2.4 or 2.0, rather than 1.1  $\mu$ W m<sup>-3</sup>), lead to a hotter Himalayan prism (reaching a difference in places of 200°C), but only minor differences beneath southern Tibet, in either the relict Indian lower crust or Tibet-derived material. The difference between our models for western and eastern Dharwar (Figure 5) shows that the thermal structure of the Himalayas is highly dependent on the distribution of radiogenic heating in the upper crust and as such is likely to be variable along strike and through time for the orogen, but that the deeper thermal structure of the lower crust and uppermost mantle is relatively insensitive to variations in shallow upper crustal radiogenic heating. Although beyond the scope of this study, we note the potential for shallow thermochronology from the Himalayas, combined with more advanced kinematics within the Himalayan prism, to further constraint the shallow thermal structure and



# Geochemistry, Geophysics, Geosystems



**Figure 3.** Preferred thermal model. (a) Model setup, kinematics, and boundary conditions. (b) Thermal model at the time when break-off is implemented (17.5 Ma). Note the change in scale from panel (a). (c) Thermal model at the present day. (d) Temperature profiles through time for a point 265 km north of the Himalayan front at the present day, along with relevant *P*-*T* constraints. Dashed boxes around xenolith data represent uncertainties. Distances back through time are corrected for the removal of material by erosion in the Himalayas. Dashed gray line shows the mantle solidus. (e) As in panel (d), but for a point 330 km north of the Himalayan front at the present day. Black area shows the permissible *P*-*T* range for xenoliths at Sailipu, but, as discussed in the text, these lack age constraints and need not be matched by our thermal model. (f) As in panel (d), but for a point 500 km north of the Himalayan front at the present day. On panels (d)–(f), ABQT constraints are shown on the profiles that most closely represent their observed location.

determine the appropriateness of these these two input models (e.g., Herman et al., 2010; Robert et al., 2011). From this point on, we show results only using the western Dharwar model.

In solving Equation 1, heat capacity, density, and thermal conductivity are all treated as temperature-dependent variables. We use experimentally derived parameterizations (see Figure S2) for





**Figure 4.** Input temperature profiles for Indian-derived lithosphere using four crustal heat production models. Brown line indicates the Moho depth. Gray line shows the mantle adiabat, for a mantle potential temperature of 1,315°C. Black dots show *P*-*T* estimates for xenoliths from Wajrakarur (Roy & Mareschal, 2011). Colored lines show steady-state geotherms for four separate crustal heat production models. Numbers in brackets indicate surface heat flux.

the temperature dependence of granodiorite (upper crust Miao et al., 2014), garnet-amphibolite (middle and lower crust; Miao et al., 2014), and olivine (mantle; McKenzie et al., 2005).

For the initial temperature structure of the Tibetan overthrust, we use a linear geothermal gradient based on fitting crustal xenolith data from Taipinghu in northern Tibet (Figure 2h, Hacker et al., 2000), although we note that the location of these xenoliths within northern Tibet places them beyond the northward extent of our model. These xenoliths constrain the temperature beneath the northern plateau interior between 30- and 50-km depths. A linear geotherm fit through these data also matches the estimated regional Curie depths from satellite magnetic data across the whole plateau, putting the 550°C isotherm at ~15 km. Similarly, the seismogenic thickness of the plateau interior, along with geodetic constraints on the locking depth of major intraplateau faults, suggests the seismic-aseismic transition at 10-12 km, also linked to the 350°C isotherm. All of these constraints are based on recent (<5 Ma) proxies and may not apply equally to the plateau back through time. We also note that a simple linear geotherm is unlikely to correctly represent the true thermal structure of north Tibetan crust. However, as will be discussed below, our more-complex thermal models produce a crustal thermal profile in central/northern Tibet equally able to fit the xenoliths data from Taipinghu, along with shallow geophysical constraints (see Figure 3f). Other than their use in constraining a crustal geotherm for northern Tibet and verifying that this does not vary significantly in the northern sections of our model, we do not use the Taipinghu xenoliths otherwise.

Few constraints exist on the distribution of radiogenic heating in the non-Indian-derived component of Tibetan crust. Much of this material was sourced from the various Cenozoic collision fronts and arc

systems that accreted onto the southern edge of Asia during the closure of the Tethys, prior to contact with continental India at ~55–50 Ma (e.g., Aitchison et al., 2007; Dewey et al., 1988; Zhu et al., 2013). As such, we can reasonably expect that this material will have been progressively enriched in incompatible radiogenic elements (including U and Th) as a result of arc magmatism. It is therefore likely that the average radiogenic heat production of Tibetan crustal material is higher than that of depleted continental lower crust, but the degree to which this enrichment has taken place and its distribution within the plateau crust are unconstrained. We hence adopt a model in which radiogenic heat production is uniform throughout the Tibetan crust and trial a series of models where the average heat production span a range of typical values of continental upper crust from 1.5 to  $2.5 \,\mu\text{W m}^{-3}$  (Jaupart & Mareschal, 2003; Rudnick et al., 1998; results shown in Figure S3). The model shown in Figure 3 uses a value of  $2.0 \,\mu\text{W m}^{-3}$ . As expected, increased internal heat production within the Tibetan overthrust leads to higher temperatures in the upper crust and midcrust of the plateau but has little effect on the deeper thermal structure within underthrust India (Figure S3).

Convergence rates in our model are based on a discretized version of the India-Asia convergence rates of (Copley et al., 2011) (see Figure 7a). These rates, however, represent the total convergence between peninsula India and stable Eurasia, only part of which is accommodated by the Himalayas. Constraints from modern geodesy (Ader et al., 2012), fluvial morphology (Lavé & Avouac, 2000), and low-temperature thermochronology (Herman et al., 2010) in the Himalayas show that convergence between northern India and central Tibet presently accommodates roughly half the total convergence rate, although this partitioning in unlikely to have been constant through time—at the present, significant shortening is accommodated in the distributed deformation belts north of Tibet (the Tien Shan, Altai, Qilian Shan, etc.), but many of these did not become active until the later Miocene/Pliocene (e.g., Bullen et al., 2001; Lease et al., 2011; Molnar & Stock, 2009). Additionally, the removal of Indian material from beneath the plateau during the early



# Geochemistry, Geophysics, Geosystems



Distance (km)

**Figure 5.** Final-stage thermal models using the four different input crustal heat production models shown in Figure 4. All other parameters are as in Figure 3 and discussed in the text. Left column shows thermal structure, and right column shows temperature differences relative to the model shown in Figure 3c. The top row shows the model from Figure 3. The second row shows a model using the crustal heat production values from eastern Dharwar from Roy and Mareschal (2011). The third row shows a model using a modified set of values for eastern Dharwar with a less-depleted lower crust. The fourth row shows a model using the heat production values from the Bundelkhand craton in northern India (Podugu et al., 2017).

Miocene likely resulted in an increase in the rates of underthrusting, as forces acting to inhibit convergence decreased, due to higher temperatures and lower viscosities beneath the plateau. In the model shown in Figure 3, we include a 50% increase in the proportion of convergence accommodated along the Himalayas at the time of break-off (to 75% of the total convergence), tapered linearly to the present-day convergence rate (50% of the total). Noting that this partitioning is unconstrained, we test a variety of temporal variations in kinematics, as summarized in Figures 7 and S4 and discussed below.

The convergence rate between northern India and central Tibet is subdivided in our modeling approach into a rate of underthrusting (i.e., the motion of Indian material relative to the Himalayan front), and a rate of overthrusting (the motion of Tibetan material relative to the Himalayan front), with overthrusting countered by the removal of material via erosion across the Himalayas. In all our models, we use the present-day partitioning between these two components of ~80% underthrusting to ~20% overthrusting (Bollinger et al., 2004; Herman et al., 2010).

We include the accretion of upper-crustal material from the top of the incoming Indian plate into the Himalayan prism. The amount of material accreted is constrained by thermochronological data from the Himalayas (Bollinger et al., 2004; Herman et al., 2010) and from seismic images that detect an intracrustal layer in southern Tibet inferred to be the contact between underthrust Indian lower crust and Asian-derived material (J. Nábělek et al., 2009; Sheehan et al., 2014). Approximately half (15–20 km) of Indian crustal material is thought to be transferred from the top of the Indian plate into the Himalayan prism. While detailed modeling of the shallow thermal structure of the Himalayas indicates that accretion potentially takes place in a concentrated zone beneath the prism (Herman et al., 2010), leading to complex intraprism kinematics, such complexity has little effect on the deeper thermal structure, and so we instead model this accretion as taking place uniformly over a 250-km-wide zone where the Indian plate dips beneath southern Tibet. The model shown in Figure 3 uses 15 km of accretion from the top of the Indian plate into the Himalayas. As shown in Figure 6, varying the amount between 10 and 20 km has a major effect on the thermal structure of the Himalayan prism, with increased accretion leading to a significantly hotter midcrust and Himalaya. Varying the amount of





**Figure 6.** Thermal models where different thicknesses of the Indian upper crust are transferred into the Himalayan prism. All other parameters are as in Figure 3, and discussed in the text. Tibetan crustal thickness is maintained at 75 km, with differing amounts of accretion compensated for by varying the thickness of the Tibetan overthrust. Left-hand column shown thermal structure, right-hand column shows temperature differences relative to the model in Figure 3.

accretion within the range of possible values leads to the second largest variation in the absolute values of the temperature structure beneath southern Tibet out of any of our model parameters, after the assumed structure for incoming Indian lithosphere. The effect on the Tibetan midcrust comes partly as a result of increased thickness of the Tibetan overthrust (required to maintain the same total crustal thickness beneath the plateau), and partly from the increased removal of the coldest sections of the incoming Indian crust. Overall, however, while absolute temperature values change, there is little effect on the general shape of the thermal structure.

For the purposes of the simple models presented here, where material is removed and instantaneously replaced by asthenosphere at a temperature given by the mantle adiabat for a potential temperature of 1,315°C, the mechanism of removal (convective delamination, slab break-off, or the progressive cascade of lithospheric material into the deeper mantle) makes minimal difference to the model outputs, provided that the rate at which removal takes place was fast in comparison to the diffusive cooling time of the material involved (in this case  $\leq$ 5 Myr). More critical is the depth at which removal takes place, the region over which it occurs, whether it removes significant crustal material from the plateau interior, and the time at which this removal occurs.

The timing of the removal of material from the plateau interior ( $t_b$ ) can be constrained by the timing of widespread magmatic activity across the central plateau from ~30 to 10 Ma (Figures 1d and 2c). As discussed elsewhere (Leary et al., 2016; Webb et al., 2017), the removal of material was probably diachronous and may have propagated across the plateau. Similarly, the southward extent of the zone of removal ( $x_b$ ) can be constrained by the spatial distribution of volcanic rocks (Figure 2d), although these locations must be corrected for the southwards motion relative to the Himalayan front since emplacement. Mantle-derived volcanism predominantly cuts off ~250 km north of the Himalayan front (although a small number of instances occur further south) and ended by ~12 Ma. Ascent times for erupted magmas are unknown but are unlikely to be >1 Myr (Mutch et al., 2019). We hence determine that mantle melting must have been possible into the later





**Figure 7.** Convergence rates and northwards extension of underthrust India for a set of kinematic scenarios. Solid lines show acceptable scenarios match or exceed the minimum inferred distance of underthrust Indian lithosphere (gray band; 500 km). Dashed lines show models that fail to meet this distance. The shaded area indicates the requirement for ultrapotassic volcanism in southern Tibet in the later Miocene. Dotted lines show models that fail to meet this criteria. Miocene within 300 km of the Himalayan front. The model shown in Figure 3 uses values of  $t_b = 17.5$  Ma and  $x_b = 250$  km. In Figures S5 and S6, we show that the effect on the model thermal structure of varying these values in the range  $12.5 < t_b < 17.5$  Ma and  $200 < x_b < 350$  km.

We also note the uncertainty in the structural geometry of the collision zone prior to removal, and the potential for there to have been more than one phase of material removal. In both of these cases, such complexity would most probably predominantly influence the crustal geotherm in Tibet used as the northern boundary condition, has little impact on the thermal structure from the final phase of removal onwards, with the principal alteration being in the temperatures of the shallow crustal material within the plateau.

The *P*-*T* values for the the Namring xenoliths provide the major constraints on the depth to which removal takes place, resulting in the presence of hot mantle material to shallow depths. In the model shown in Figure 3, we include the removal of the lower 10 km of the underthrust Indian crust, allowing mantle material at near-asthenospheric temperatures to reach 65 km beneath the plateau, allowing the fit to the xenolith data shown in Figure 3d. However, whether that is the result of the removal of any crustal material or instead a product of a lesser crustal thickness of the plateau in the Miocene is unconstrained by our modeling.

## 6. Discussion

Figure 3 shows one thermal model for southern Tibet consistent with all available required constraints. As discussed, in this model, all Indian mantle and the lower 10 km of the Indian crust north of 250 km north of the Himalayan front are removed at 17.5 Ma and replaced with adiabatic mantle (Figure 3b). The top 15 km of the Indian crust is transferred into the Himalayan prism during under-thrusting. Figure 3c shows the modeled thermal structure for the present day, for comparison with geophysical observables. Figures 3e and 3f show a series of temperature profiles for three present-day distances from the plateau margin, illustrating the temperature evolution through time, for comparison with geochemical and geophysical proxies. We will use this model as the basis for the following discussion, but emphasize that it is only one of a range of possible models which are capable of fitting much, if not all, of the available temperature information beneath southern Tibet.

The overall "shape" of the thermal structure shown in Figure 3 is common to all our models and to previous studies. Figure 3 represents one scenario capable of appropriately fitting available constraints on the absolute temperature values. The variability in absolute temperature values shown in Figures 5 and 6 serves to illustrate the degree to which certain areas of the model are unknown—in particular, the temperatures present in the Tibetan midcrustal section, around 30- to 40-km depth. The only direct constraints in temperatures at this depth available for southern Tibet are those from estimates of the ABQT, and the indirect impact that changes in midcrustal temperatures have on the better-constrained temperatures in the shallow crust, and near the Moho. As such, the midcrust may vary by up to 200°C from the values shown in Figure 3, while still being able to fit much of the data. Such variation could result most plausibly from variations in accretion (Figure 6), input thermal structure (Figure 5), or the internal heat production to Tibetan material (Figure S3). Temperatures in the shallow crust and around the Moho are much better constrains from the geophysical and geochemical data available. We assess that,



within the framework of our models as described above, these are unlikely to vary by more than 50–100°C, without leading to significant discrepancies in matching the constraints shown in Figure 2.

#### 6.1. Comparison to Previous Models of the Thermal Structure of Southern Tibet

Several previous studies have constructed numerical models for the thermal structure of southern Tibet (e.g., Bollinger et al., 2006; Craig et al., 2012; Henry et al., 1997; Herman et al., 2010; Hétenyi et al., 2007; P. Nábělek & Nábělek, 2014; Wang et al., 2013; Whipp Jr. et al., 2007). The geometry of these models are all, to first order, similar, with India underthrusting beneath the Himalayas, and, in most cases where models extend sufficiently far northwards, flattening out beneath southern Tibet. Depending on the focus of the study in question (typically shallow thermochronometry in the Himalayas, Bollinger et al., 2006; Henry et al., 1997; Herman et al., 2010; Whipp Jr. et al., 2007; or deeper thermal structure beneath the plateau, Craig et al., 2012; Hétenyi et al., 2007; Wang et al., 2013), slightly different approaches to the geometry of the main interface beneath the Himalayas and the kinematics of crustal accretion during underthrusting are incorporated, although, as we note above, these have little impact on the deeper thermal structure. Similarly, as discussed above, the inclusion of a shear (or strain) heating term in Herman et al. (2010) and P. Nábělek and Nábělek (2014), while potentially playing a role in defining the shallow thermal structure of the Himalayas, has little effect on the thermal structure on Indian crustal and mantle beneath southern Tibet.

All of these prior models focus on the present-day thermal structure and, in many case, assume it is in steady state. As a result, the predictions of our model with regards to the past thermal history are not included in this comparison.

The most significant differences between past models, and those that we present here, arise from variations in the thermal structure of incoming Indian lithosphere, and the model kinematics, particularly the rate of underthrusting and the amount of material removed from the top of the Indian plate. As discussed above, the input thermal structures used here are built on data from several regions of peninsula India south of the collision zone. That used in the calculations in Figure 3 is based on crustal heat production data and heat flow from western Dharwar, along with mantle xenoliths from beneath the Dharwar craton. This differs from previous input models, in part due to the inclusion of temperature-dependent material properties and in part due to different assumptions about crustal radiogenic heat production.

The model of P. Nábělek and Nábělek (2014) uses an Indian thermal structure with 100-km-thick lithosphere, and is therefore significantly hotter than any of our models or the other models considered. As a result, while the same overall shape emerges, with a tongue of colder Indian material thrust beneath the plateau, underthrust material does not persist and <700°C beyond ~200 km from the Himalayan front. Such thin, hot lithosphere from incoming India, while potentially valid for material already thrust beneath the plateau, is inconsistent with measurements of the thermal structure of the cratonic areas of the Indian peninsula, or with its rheology, and unable to fit the majority of the constraints we describe in sections 2 and 3. Similar issues apply to the model of Henry et al. (1997), which also uses a 100-km-thick India lithosphere, as well as relatively high ( $2.5 \,\mu\text{W} \,\text{m}^{-3}$ ) upper crustal heat production, leading to a very hot incoming Indian lithosphere.

The models of Bollinger et al. (2006), Hétenyi et al. (2007), Herman et al. (2010), and Whipp Jr. et al. (2007) all use a restricted model domain, typically only including the upper ~80 km of the collision zone, and as such are not appropriate for considering the deeper thermal structure, close to the lower boundary condition at the base of their models (60- to 80-km depth). However, for the crustal section, Bollinger et al. (2006), Whipp Jr. et al. (2007), and Herman et al. (2010) use extremely cold geotherms, most closely comparable to our model using a geotherm from Bundelkhand (Figure 5, lowermost panels). While that of Herman et al. (2010) is determined through the inversion of thermochronological data from the Himalayas, as discussed above, a geotherm this cold for incoming India is highly unlikely to be representative of much of incoming Indian lithosphere, as it would produce an extremely strong incoming plate, seismogenic to depths of  $\geq 60 \text{ km}$ , inconsistent with estimates of the seismogenic and elastic thickness in India. The thermal structure from either model would (unless limited by the removal of material beneath the plateau) lead to the persistence of temperatures  $\leq 600^{\circ}$ C significantly further beneath the plateau that in all our models, bar the Bundelkhand model from Figure 5, and would also lead to a decrease in midcrustal temperatures, unless countered by even higher radiogenic heating in Tibetan material than used in our models.

Most similar to our models, especially that shown in Figure 3 are those of Hétenyi et al. (2007) and Craig et al. (2012). Hétenyi et al. (2007) use a similar input geothermal for incoming India to the models for the Dharwar craton we use (Figures 4 and 5), but, as with Bollinger et al. (2006) and Herman et al. (2010), they use a restricted model domain unsuitable for studying the deeper thermal structure of southern Tibet, with a constant heat flow boundary condition imposed at 80 km depth. In contrast, the models presented here demonstrate that the temperature and heat flux at a depth of 80 km vary in space and time. The models presented here are broadly similar to that shown in Craig et al. (2012), updated with improved data on crustal radiogenic heat production, model kinematics, and the thermal parameters of crustal rocks, all described above.

#### 6.2. Fit to Temperature Proxies Beneath Tibet

Our model is capable of matching the distance distribution of deeper seismicity beneath India and Tibet and of shallow seismicity across the plateau, with an aseismic region between, based on the expected temperature cut-offs discussed above. The emergence of the region of inefficient head-wave propagation beyond  $\sim$ 500 km north of the Himalayan front fits with the transition to a negative temperature gradient across the Moho at this distance (right-hand side, Figures 3c and 3f), although the nature and origin of the lower under most of this region is not encompassed by our model. Similarly, the development of a consistent anisotropic fabric at a similar distance fits with the progressive northwards heating and weakening of the crustal material, and the development of significant ductile deformation within the midcrust, as temperatures exceed 750°C, although we note that without knowledge of the mineralogy and stress state, it is not possible to make a quantitative comparison between our model results and the anisotropy data.

As the present-day temperature profiles (red lines in Figures 3d–3f) show, our model does a good job of fitting the estimated depth of the Curie isotherm across Tibet (Alsdorf & Nelson, 1999), fitting to within  $\sim$ 50°C or a 1-km error in depth. Fits for the ABQT depths for southern Tibet from Mechie et al. (2004) (Figures 3e and 3f) similarly are well explained with our models. Fitting the ABQT depth of Sheehan et al. (2014) in the Himalayan midcrust is more problematic (see Figures 3d and 3e), with the implied high midcrustal temperatures failing to match our models by  $\sim$ 250°C, or  $\sim$ 100°C and a 5-km decrease in depth. Fitting the pressure and temperature estimates from the ABQT at this location requires either extreme amounts of radiogenic heat production within the plateau, implausibly rapid erosion and exhumation of material through the Himalayas (and therefore likely incompatible with shallow thermochronology Herman et al., 2010), or potentially the inclusion of shear heating along the interface in the Himalayan prism (P. Nábělek & Nábělek, 2014).

We now consider the fit to geochemical data across southern Tibet. In comparing our thermal models to geochemical data (Figures 3d-3f), we recalculate the position of geochemical samples back through time based on the movement of material within the plateau toward India as a result of the erosive removal of material through the Himalayan front, using the integrated erosive component of the advective velocity field between the present and the time of emplacement of each sample. Profiles prior to 0 Ma on Figures 3e and 3f therefore do not represent the distance quoted, but the greater distance representative for that point at each period in time.

Our model fits the constraints derived from north Tibetan xenoliths from Taipinghu (Figure 3f), although the linear geotherm assumed for the northern boundary condition is not entirely appropriate. As a result of the removal of material from beneath the plateau at 17.5 Ma, our model reproduces the distribution and latest occurrence of mantle melting beneath the plateau. As shown by the 15-Ma profile on Figure 3d, our model also matches the *P*-*T*-*t* estimates from the Namring xenoliths. We note that, in our modeling approach, matching both the present-day crustal thickness of the plateau, the *P* and *T* values for these xenoliths and a mantle source, requires the removal of the lowermost Indian crust during break-off, but this could equally be accommodated by having a thinner crust beneath the plateau prior to the removal of mantle material, and neither possibility can be discounted. A removal age of 17.5 Ma is best able to fit the Namring xenolith data, but this remains uncertain to within a few million years. The time at which this removal of material occurs has little effect on the final thermal structure, due to subsequent re-underthrusting (see Figure S5), and trades off with the postremoval convergence rate (see Figure 7). As discussed earlier, the lack of age constraints for the *P*-*T* estimates from xenoliths at Sailipu and Yibuchaka, along with the uncertain nature of pressure estimates based purely on the lack of garnet, mean they place only limited constraints on the deeper thermal structure of the plateau. However, we note that for for Sailipu (Figure 3e), our model would come close to replicating the required *P*-*T* conditions shortly prior to their emplacement near the surface at 17 Ma. However, a complete match to the Sailipu or Yibuchaka xeno-lith *P*/*T* ranges (shown in Figure 3e) would require the removal of a greater thickness of the lower crust than in the model shown in Figure 3, in order to get the required temperatures at a sufficiently shallow depth to be above the garnet-spinel transition.

Overall, the model shown in Figure 3 is capable of matching both the majority of the geophysical constraints available on the present-day structure of the plateau and the geochemical constraints on the plateau evolution in the early-mid Miocene. The difficulty of matching both the geophysical data on the modern plateau and the geochemical data for the prior thermal structure of the plateau requires two critical tectonic events: the removal of significant volumes of mantle material to depths close to the Moho either at or continuously until the mid Miocene and the re-underthrusting of colder Indian material beneath southern Tibet since that removal. To fit the modern-day geophysical structure of southern Tibet, this underthrusting is required to emplace a cold, subhorizontal Indian slab beneath southern Tibet to a minimum of  $\sim$ 500 km north of the Himalayan front. Given that the tip of this slab will also be heated by lateral diffusion at its end, its true extent is likely greater.

For the tip of the underthrust Indian lithosphere to reach such distances following removal of lithosphere beyond ~250 km north of the Himalayan front, while also having break-off at a sufficient distance and time to match the occurrence of high Mg# volcanism, requires that the average rate of underthrusting (i.e., the motion of India relative to central Tibet) since break-off was faster than it is at present, although the rate of underthrusting trades off against the time and southernmost extent of break-off. Figure 7 plots convergence rates and penetration distances of underthrust Indian material for a series of convergence models, demonstrating that models (i) with removal extending farther south than that shown in Figure 3, (ii) ending later than in Figure 3, or (iii) without acceleration after removal would fail to match the present-day northwards extent of India beneath southern and central Tibet. This is subject to the caveat that our model assumes uniformity along strike in terms of both the break-off distance and the penetration distance of present-day India beneath the plateau, and some variability in these values could potentially reduce the required slow down in convergence rates, while still fitting available data constraints.

Underthrusting since the late Miocene at a rate greater than the current rate at which Indian material is thrust beneath Tibet is conceptually in alignment with the reduction in resistive force likely to result from the removal of stalled Indian mantle beneath central/southern Tibet. Additionally, deformation in the distal regions of the India-Asia collision zone (e.g., Tien Shan, Qilian Shan, Altai, and Baikal Rift) either initiated or increased over the period following break-off (Bullen et al., 2001; Lease et al., 2011; Molnar & Stock, 2009; Petit & Déverchère, 2006). Given that the total convergence rate varied relatively little over this time period (Copley et al., 2010; Molnar & Stock, 2009), this deformation requires an implicit decrease in the rates of deformation along the Himalayan front over the last ~15 Ma.

#### 7. Conclusion

Simple models for the temperature evolution of southern Tibet are able to reconcile all available geophysical and geochemical information on the deep thermal structure of the plateau through time. Three key features from these models emerge. First, present-day observations require the presence of a cold Indian plate emplaced beneath southern Tibet by large-scale underthrusting. Second, geochemical evidence requires the removal of Indian-derived mantle (and potentially lower-most crust) from beneath the plateau during the early Miocene. Our models are unable to resolve whether crustal removal is also required, or if crustal thickness of the plateau has varied through time. Lastly, to match both the southwards extent of Miocene mantle-derived volcanism, and the present-day geophysical structure of the plateau, the average rate since the middle Miocene at which India underthrust Tibet must have been faster than it is at present.

#### Data Availability Statement

All data used in this study have previously been published and are available as described in the text.



#### Acknowledgments

TJC was supported in this work by the Royal Society under URF\R1\180088 TJC and AC were also supported through COMET, the UK Natural Environment Research Council's Centre for the Observation and Modelling of Earthquakes, Volcanoes, and Tectonics. BRH was supported by NSF-EAR 1551054. PBK was supported in part by the Arthur D. Storke Chair at Columbia University. We thank Matthew Agius for making his compilation of splitting measurements available and Owen Weller for detailed comments on an initial manuscript, and we thank Dan McKenzie. Helen Williams, and David Ferguson for useful discussions. We thank the editor, associate editor, and two anonymous reviewers for the their comments on our manuscript.

#### References

- Ader, T., Avouac, J.-P., Liu-Zeng, J., Lyon-Caen, H., Bollinger, L., Galetzka, J., et al. (2012). Convergence rate across the Nepal Himalaya and interseismic coupling on the Main Himalayan Thrust: Implications for seismic hazard. *Journal of Geophysical Research*, *117*, B04403. https://doi.org/10.1029/2011JB009071
- Agius, M. R., & Lebedev, S. (2017). Complex, multilayered azimuthal anisotropy beneath Tibet: Evidence for co-existing channel flow and pure-shear crustal thickening. *Geophysical Journal International*, 210, 1823–1844. https://doi.org/10.1093/gji/ggx266
- Ainscoe, E. A., Elliott, J. R., Copley, A., Craig, T. J., Li, T., Parsons, B. E., & Walker, R. T. (2017). Blind thrusting, surface folding, and the development of geological structure in the M<sub>w</sub> 6.3 2015 Pishan (China) earthquake. Journal of Geophysical Research: Solid Earth, 122, 9359–9382. https://doi.org/10.1002/2017JB014268

- Alsdorf, D., & Nelson, D. (1999). Tibetan satellite magnetic low: Evidence for widespread melt in the Tibetan crust? *Geology*, 27, 943–946. https://doi.org/10.1130/0091-7613(1999)027<0943:TSMLEF>2.3.CO;2
- Alvizuri, C., & Hetényi, G. (2019). Source mechanism of a lower crust earthquake beneath the Himalayas and its possible relation to metamorphism. *Tectonophysics*, 769, 128,153. https://doi.org/10.1016/j.tecto.2019.06.023
- Angel, R. J., Alvaro, M., Miletich, R., & Nestola, F. (2017). A simple and generalised P-T-V EoS for continuous phase transitions, implemented in EosFit and applied to quartz. *Contributions to Mineralogy and Petrology*, 172, 29. https://doi.org/10.1007/s00410-017-1349

Barron, J., & Priestley, K. (2009). Observations of frequency-dependent S<sub>n</sub> propagation in Northern Tibet. *Geophysical Journal International*, 179, 475–488. https://doi.org/10.1111/j.1365-246X.2009.04318.x

Boettcher, M. S., Hirth, G., & Evans, B. (2007). Olivine friction at the base of oceanic seismogenic zones. *Journal of Geophysical Research*, 112, B01205. https://doi.org/10.1029/2006JB004301

- Bollinger, L., Avouac, J.-P., Beyssac, O., Catlos, E. J., Harrison, T. M., Grove, M., et al. (2004). Thermal structure and exhumation history of the Lesser Himalaya in central Nepal. *Tectonics*, 23, TC5015. https://doi.org/10.1029/2003TC001564
- Bollinger, L., Henry, P., & Avouac, J.-P. (2006). Mountain building in the Nepal Himlaya: Thermal and kinematic model. *Earth and Planetary Science Letters*, 244, 58–71. https://doi.org/10.1016/j.epsl.2006.01.045
- Bullen, M. E., Burbank, D. W., Garver, J. I., & Abdrakhmatov, K. Y. (2001). Late Cenozoic tectonic evolution of the northwestern Tien Shan: New age estimates for the initiation of mountain building. GSA Bulletin, 113, 154–1559. https://doi.org/10.1130/0016-7606(2001)113<1544.LCTEOT>2.0.CO;2

Carrapa, B., Orme, D. A., DeCelles, P. G., Kapp, P., Cosca, M. A., & Waldrip, R. (2014). Miocene burial and exhumation of the India-Asia collision zone in southern Tibet: Response to slab dynamics and erosion. *Geology*, *42*, 443–446. https://doi.org/10.1130/G35350.1

Cattin, R., Martelet, G., Henry, P., Avouac, J. P., Diament, M., & Shakya, T. R. (2001). Gravity anomalies, crustal structure and thermo-mechanical support of the Himalaya of Central Nepal. *Geophysical Journal International*, 147, 381–392.

Chan, G. H.-N., Waters, D. J., Searle, M. P., Aitchison, J. C., Horstwood, M. S. A., Crowley, Q., et al. (2009). Probing the basement of southern Tibet: Evidence from crustal xenoliths entrained in a Miocene utrapotassic dyke. *Journal of the Geological Society of London*, 166, 45–52. https://doi.org/10.1144/0016-76492007-145

Chen, W.-P., & Molnar, P. (1983). Focal depths of intracontinental and intraplate earthquakes and their implications for the thermal and mechanical properties of the lithosphere. *Journal of Geophysical Research*, *88*, 4183–4214.

- Chen, M., Niu, F., Tromp, J., Lenardic, A., Lee, C.-T. A., Cao, W., & Ribeiro, J. (2017). Lithospheric foundering and underthrusting imaged beneath Tibet. *Nature Communications*, *8*, 15,659. https://doi.org/10.1038/ncomms15659
- Chung, S.-L., Chu, M.-F., Zhang, Y., Xie, Y., Lo, C.-H., Lee, T.-Y., et al. (2005). Tibetan tectonic evolution inferred from spatial and temporal variations in post-collisional magmatism. *Earth-Sciences Reviews*, 68, 173–196. https://doi.org/10.1016/j.earscirev.2004.05.001
- Copley, A., Avouac, J.-P., & Royer, J.-Y. (2010). India-Asia collision and the Cenozoic slowdown of the Indian plate: Implications for the forces driving plate motions. *Journal of Geophysical Research*, *115*, B03410. https://doi.org/10.1029/2009JB006634
- Copley, A., Avouac, J.-P., & Wernicke, B. P. (2011). Evidence for mechanical coupling and strong Indian lower crust beneath southern Tibet. *Nature*, 472, 79–81. https://doi.org/10.1038/nature09926

Craig, T. J., Copley, A., & Jackson, J. (2012). Thermal and tectonic consequences of India underthrusting Tibet. Earth and Planetary Science Letters, 353-354, 231–239. https://doi.org/10.1016/j.epsl.2012.07.010

Craig, T. J., Copley, A., & Jackson, J. (2014). A reassessment of outer-rise seismicity and its implications for the mechanics of oceanic lithosphere. *Geophysical Journal International*, 197, 63–89. https://doi.org/10.1093/gji/ggu013

DeCelles, P. G., Robinson, D. M., & Zandt, G. (2002). Implications of shortening in the Himalayan fold-thrust belt for uplift of the Tibetan Plateau. *Tectonics*, 21, 1062. https://doi.org/10.1029/2001TC001322

Diehl, T., Singer, J., Hetényi, G., Grujic, D., Clinton, J., Giardini, D., Kissling, E., & GANSSER Working Group (2017). Seismotectonics of Bhutan: Evidence for segmentation of the Eastern Himalayas and link to foreland deformation. *Earth and Planetary Science Letters*, 471, 54–64. https://doi.org/10.1016/j.epsl.2017.04.038

- Ding, L., Kapp, P., Yue, Y., & Lai, Q. (2007). Postcollisional calc-alkaline lavas and xenoliths from the southern Qiangtang terrane, central Tibet. *Earth and Planetary Science Letters*, 254, 28–38. https://doi.org/10.1016/j.epsl.2006.11.019
- Elliott, J. R., Walters, R. J., England, P. C., Jackson, J. A., Li, Z., & Parsons, B. (2010). Extension on the Tibetan plateau: Recent normal faulting measured by InSAR and body wave seismology. *Geophysical Journal International*, 183, 503–535. https://doi.org/10.1111/j.1365-246X.2010.04754.x

Gibbons, S., & Kvaerna, T. (2017). Illuminating the seismicity pattern of the October 8, 2005, M = 7.6 Kashmir earthquake aftershocks. *Physics of the Earth and Planetary Interiors*, 270, 1–8. https://doi.org/10.1016/j.pep.2017.06.008

Gilligan, A., & Priestley, K. (2018). Lateral variations in the crustal structure of the Indo-Eurasian collision zone. *Geophysical Journal International*, 214, 975–989. https://doi.org/10.1093/gji/ggy172

- Godin, L., Parrish, R. R., Brown, R. K., & Hodges, K. V. (2001). Crustal thickening leading to exhumation of the Himalayan Metamorphic core of central Nepal: Insight from U-Pb Geochronology and <sup>40</sup>Ar/<sup>39</sup>Ar Thermochronology. *Tectonics*, 20, 729–747. https://doi.org/ 10.1029/2000TC001204
- Guillot, S., Garzanti, E., Baratoux, D., Marquer, D., Mah'eo, G., & de Sigoyer, J. (2003). Reconstructing the total shortening history of the NW Himalaya. *Geochemistry, Geophysics, Geosystems*, 4(7), 1064. https://doi.org/10.1029/2002GC000484

Aitchison, J. C., Ali, J. R., & Davies, A. M. (2007). When and where did India and Asia collide? Journal of Geophysical Research, 112, B05423. https://doi.org/10.1029/2006JB004706

Dewey, J. F., Shackleton, R. M., Chengfa, C., & Yiyin, S. (1988). The tectonic evolution of the Tibetan Plateau. Philosophical Transactions of the the Royal Society A, 327, 379–413. https://doi.org/10.1098/rsta.1988.0135



- Hacker, B. R., Gnos, E., Ratschbacher, L., Grove, M., McWilliams, M., Sobolev, S. V., et al. (2000). Hot and dry deep crustal xenoliths from Tibet. Science, 287, 2463–2466. https://doi.org/10.1126/science.287.5462.2463
- Henry, P., Le Pichon, X., & Goffé, B. (1997). Kinematic, thermal and petrological model of the Himalayas: Constraints related to metamorphism within the underthrust Indian crust and topographic elevation. *Tectonophysics*, 273, 31–56.
- Herman, F., Copeland, P., Avouac, J.-P., Bollinger, L., Mahéo, G., Le Fort, P., et al. (2010). Exhumation, crustal deformation, and thermal structure of the Nepal Himalaya derived from the inversion of thermochronological and thermobarometric data and modeling of the topography. *Journal of Geophysical Research*, 115, B06407. https://doi.org/10.1029/2008JB006126
- Hétenyi, G., Cattin, R., Brunet, F., Bollinger, L., Vergne, J., Nábělek, J. L., & Diament, M. (2007). Density distribution of the India plate beneath the Tibetan plateau: Geophysical and petrological constraints on the kinetics of lower-crustal eclogitization. *Earth and Planetary Science Letters*, 264, 226–244. https://doi.org/10.1016/j.epsl.2007.09.036
- Huang, G. D., Roecker, S. W., Levin, V., Wang, H., & Li, Z. (2017). Dynamics on intracontinental convergence between the western Tarim basin and central Tien Shan constrained by centroid moment tensors of regional earthquakes. *Geophysical Journal International*, 208, 561–576. https://doi.org/10.1093/gji/ggw415
- Hubbard, J., Almeida, R., Foster, A., Sapkota, S. N., & Bürgi, P. (2016). Structural segmentation controlled the 2015 M<sub>w</sub> 7.8 Gorkha earthquake rupture in Nepal. *Geology*, 44, 639–642. https://doi.org/10.1130/G38077.1

Ingleby, T., Wright, T. J., Hooper, A., Craig, T. J., & Elliott, J. R. (2020). Constraints on the geometry and frictional properties of the Main Himalayan Thrust using coseismic, postseismic and interseismic deformation in Nepal. *Journal of Geophysical Research: Solid Earth*, 125, e2019JB019201. https://doi.org/10.1029/2019JB019201

International Heat Flow Commission (2011). Global Heatflow Database. http://ihfc-iugg.org/products/global-heat-flow-database

- Jackson, J., McKenzie, D., Priestley, K., & Emmerson, B. (2008). New views on the structure and rheology of the lithosphere. *Journal of the Geological Society, London, 165*, 453–465. https://doi.org/10.1144/0016-76492007-109
- Jamtveit, B., Petley-Ragan, A., Incel, S., Dunkel, K. G., Aupart, C., Austrheim, H., et al. (2019). The effects of earthquakes and fluids on the metamorphism of the lower continental crust. *Journal of Geophysical Research: Solid Earth*, 124, 7725–7755. https://doi.org/10.1029/ 2018JB016461
- Jaupart, C., & Mareschal, J.-C. (2003). Constraints on crustal heat production from heat flow data. In Rudnick, R. (Ed.), *Treatise on Geochemistry: the Crust* (Vol.3, pp. 65–84): Elsevier.
- Kanna, N., Gupta, S., & Prakasam, K. S. (2018). Micro-seismicity and seismotectonic study in Western Himalaya-Ladakh-Karakorum using local broadband seismic data. *Tectonophysics*, 726, 100–109. https://doi.org/10.1016/j.tecto.2018.01.032
- Kufner, S.-K., Schurr, B., Sippl, C., Yuan, X., Ratschbacher, L., s/of Mohammed Akbar, A., et al. (2016). Deep India meets deep Asia: Lithospheric indentation, delamination and break-off under Pamir and Hindu Kush (Central Asia). Earth and Planetary Science Letters, 435, 171–184. https://doi.org/10.1016/j.epsl.2015.11.046

Langin, W. R., Brown, L. D., & Sandvol, E. A. (2003). Seismicity of Central Tibet from Project INDEPTH III Seismic Recordings. Bulletin of the Seismoglogical Society of America, 93, 2146–2159.

- Lavé, J., & Avouac, J. P. (2000). Active folding of fluvial terraces across the Siwaliks Hills, Himalayas of central Nepal. Journal of Geophysical Research, 105, 5735–5770.
- Leary, R., Orme, D. A., Laskowski, A. K., DeCelles, P. G., Kapp, P., Carrapa, B., & Dettinger, M. (2016). Along-strike diachroneity in deposition of the Kailas Formation in central southern Tibet: Implications for Indian slab dynamics. *Geosphere*, 12, 1198–1223. https:// doi.org/10.1130/GES01325.1
- Lease, R. O., Burbank, D. W., Clark, M. K., Farley, K. A., Zheng, D., & Zhang, H. (2011). Middle Miocene reorganisation of deformation along the northeastern Tibetan Plateau. *Geology*, 39, 359–362. https://doi.org/10.1130/G31356.1
- Li, J., & Song, X. (2018). Tearing of Indian mantle lithosphere from high-resolution seismic images and its implications for lithosphere coupling in southern Tibet. Proceedings of the National Academy of Sciences, 115, 8296–9300. https://doi.org/10.1073/ pnas.1717258115
- Liang, X., Chen, Y., Tian, X., Chen, Y. J., Ni, J., Gallegos, A., et al. (2016). 3D imaging of subducting and fragmenting Indian continental lithosphere beneath southern and central Tibet using body-wave finite-frequency tomography. *Earth and Planetary Science Letters*, 443, 162–175. https://doi.org/10.1016/j.epsl.2016.03.029
- Liu, C.-Z., Wu, F.-Y., Chung, S.-L., & Zhao, Z.-D. (2011). Fragments of hot and metasomatized mantle lithosphere in Middle Miocene ultrapotassic lavas, southern Tibet. 2011, 39, 923–926. https://doi.org/10.1130/G32172.1

McKenzie, D., Jackson, J., & Priestley, K. (2005). Thermal structure of oceanic and continental lithosphere. Earth and Planetary Science Letters, 233, 337–349. https://doi.org/10.1016/j.epsl.2005.02.005

Mackwell, S. J. (1998). High-temperature deformation of dry diabase with application to tectonics on Venus. *Journal of Geophysical Research*, *103*, 975–984.

McNamara, D. E., & Owens, T. J. (1995). Observations of regional phase propagation across the Tibetan Plateau. Journal of Geophysical Research, 100, 22,215–22,229.

Mechie, J., Sobolev, S. V., Ratschbacher, L., Babeyko, A. Y., Bock, G., Jones, A. G., et al. (2004). Precise temperature estimation in the Tibetan crust from seismic detection of the  $\alpha$ - $\beta$  quartz transition. *Science*, *32*, 601–604. https://doi.org/10.1130/20367.1

Mendoza, M. M., Ghosh, A., & Rai, S. S. (2016). Dynamic triggering of small local earthquakes in the central Himalaya. Geophysical Research Letters, 43, 9581–9587. https://doi.org/10.1002/2016GL069969

Miao, S. Q., Li, H. P., & Chen, G. (2014). Temperature dependence of thermal diffusivity, specific heat capacity, and thermal conductivity for several types of rocks. *Journal of Thermal Analysis and Calorimetry*, *115*, 1057–1063. https://doi.org/10.1007/s10973-013-3427-2

Mitra, S., Wanchoo, S., & Priestley, K. F. (2014). Source parameters of the 1 May 2013 m<sub>b</sub> 5.7 Kishtwar earthquake: Implications for seismic hazards. Bulletin of the Seismological Society of America, 104, 1013–1019. https://doi.org/10.1785/0120130216

Molnar, P., & Chen, W.-P. (1983). Focal depths and fault plane solutions of earthquakes under the Tibetan Plateau. Journal of Geophysical Research, 88, 1180–1196.

Molnar, P., & Stock, J. M. (2009). Slowing of India's convergence with Eurasia since 20 Ma and its implications for Tibetan mantle dynamics. *Tectonics*, 28, TC3001. https://doi.org/10.1029/2008TC002271

Monsalve, G., Sheehan, A., Schulte-Pelkum, V., Rajaure, S., Pandey, M. R., & Wu, F. (2006). Seismicity and one-dimensional velocity structure of the Himalayan collisions zone: Earthquakes in the crust and upper mantle. *Journal of Geophysical Research*, 111, B10301. https://doi.org/10.1029/2005JB004062

Mugnier, J.-L., & Huyghe, P. (2006). Ganges basin geometry records a pre-15 Ma isostatic rebound of Himalaya. *Geology*, 34, 445–448. https://doi.org/10.1130/G22089.1



- Mutch, E. J. F., Maclennan, J., Holland, T. J. B., & Buisman, I. (2019). Millenial storage of near-Moho magma. Science, 19, 260–264. https:// doi.org/10.1126/science.aax4092
- Nábělek, J., Hetényi, G., Vergne, J., Sapkota, S., Kafle, B., Jiang, M., et al. (2009). Underplating in the Himalaya-Tibet collision zone revealed by the Hi-CLIMB experiment. *Science*, 325, 1371–1374. https://doi.org/10.1126/science.1167719
- Nábělek, P., & Nábělek, J. (2014). Thermal characteristics of the Main Himlayan Thrust and the Indian lower crust with implications for crustal rheology and partial melting in the Himalaya orogen. Earth and Planetary Science Letters, 395, 116–123. https://doi.org/10.1016/j. epsl.2014.03.026
- Najman, Y., Burley, S. D., Copley, A., Kelly, M. J., Pander, K., & Mishra, P. (2018). The late Eocene-early Miocene unconformities of the NW Indian intraplate basins and Himalayan foreland: A record of tectonics or mantle dynamics. *Tectonics*, 37, 3970–3985. https://doi.org/ 10.1029/2018TC005286
- Negi, S. S., Paul, A., Cesca, S., Kamal, Kriegerowski, M., Mahesh, P., & Gupta, S. (2017). Crustal velocity structure and earthquake processes of Garhwal-Kumaum Himalaya: Constraints from regional waveform inversion and array beam modeling. *Tectonophysics*, 712-713, 45–63. https://doi.org/10.1016/j.tecto.2017.05.007
- Ni, J., & Barazangi, M. (1983). High-frequency seismic wave propagation beneath the Indian Shield, Himalayan Arc, Tibetan Plateau and surrounding regions: High uppermost mantle velocities and efficient S<sub>n</sub> propagation beneath Tibet. *Geophysical Journal International*, 72, 665–689. https://doi.org/10.1111/j.1365-246X.1983.tb02826.x

Pandey, S., Yuan, X., Debayle, E., Tilmann, F., Priestley, K., & Li, X. (2015). Depth-variant azimuthal anisotropy in Tibet revealed by surface wave tomography. *Geophysical Research Letters*, 42, 4326–4334. https://doi.org/10.1002/2015GL063921

Parija, M. P., Kumar, S., Tiwari, V. M., Rao, N. P., Kumar, N., Biswal, S., & Singh, I. (2018). Microseismicity, tectonics and seismic potential in the Western Himalayan segment, NW Himalaya, India. *Journal of Asian Earth Sciences*, 159, 1–16. https://doi.org/10.1016/j. jseaes.2018.03.016

- Paul, H., Mitra, S., Bhattacharya, S. N., & Suresh, G. (2015). Active transverse faulting within underthrust Indian crust beneath the Sikkim Himalaya. *Geophysical Journal International*, 2015, 1072–1083. https://doi.org/10.1093/gji/ggv058
- Petit, C., & Déverchère, J. (2006). Structure and evolution of the Baikal rift: A synthesis. *Geochemistry, Geophysics, Geosystems*, 7, Q11016. https://doi.org/10.1029/2006GC001265
- Podugu, N., Ray, L., Singh, S. P., & Roy, S. (2017). Heat flow, heat production, and crustal temperatures in the Archaean Bundelkhand craton, north-central India: Implications for thermal regime beneath the Indian shield. *Journal of Geophysical Research: Solid Earth*, *122*, 5766–5788. https://doi.org/10.1002/2017JB014041
- Priestley, K., McKenzie, D., Debayle, E., & Pilidou, S. (2008). The African upper mantle and its relationship to tectonics and surface geology. *Geophysical Journal International*, 175, 1108–1126. https://doi.org/10.1111/j.1365-246X.2008.03951.x

Ren, Y., & Shen, Y. (2008). Finite frequency tomography in southeastern Tibet: Evidence for the causal relationship between mantle lithosphere delamination and the north-south trending rifts. *Journal of Geophysical Research*, 113, B10316. https://doi.org/10.1029/ 2008JB005615

- Replumaz, A., Capitanio, F. A., Guillot, S., Negredo, A. M., & Villaseñor, A. (2014). The coupling on Indian subduction and Asian continental tectonics. *Gondwana Research*, 26, 608–626. https://doi.org/10.1016/j.gr.2014.04.003
- Replumaz, A., Negredo, A. M., Guillot, S., & Villanseñor, A. (2010). Multiple episodes of continental subduction during India/Asia convergence: Insight from seismic tomography and tectonic reconstruction. *Tectonophysics*, 483, 125–124. https://doi.org/10.1016/j. tecto.2009.10.007
- Reynolds, K., Copley, A., & Hussain, E. (2015). Evolution and dynamics of a fold-thrust belt: The Sulaiman Range of Pakistan. *Geophysical Journal International*, 201, 683–710. https://doi.org/10.1093/gji/ggv005
- Robert, X., van der Beek, P., Braun, J., Perry, C., & Mugnier, J.-L. (2011). Control of detachment geometry on lateral variations in exhumation rates in the Himalaya: Insights from low-temperature thermochronology and numerical modeling. *Journal of Geophysical Research*, 116, B05202. https://doi.org/10.1029/2010JB007893
- Roy, S., & Mareschal, J-C. (2011). Constraints on the deep thermal structure of the Dharwar craton, India, from heat flow, shear wave velocities, and mantle xenoliths. *Journal of Geophysical Research*, *116*, B02409. https://doi.org/10.1029/2010JB007796
- Rudnick, R. L., McDonough, W. F., & O'Connell, R. J. (1998). Thermal structure, thickness and composition of continental lithosphere. *Chemical Geology*, 145, 395–411.
- Schulte-Pelkum, V., Monsalve, G., Sheehan, A. F., Shearer, P., Wu, F., & Rajaure, S. (2019). Mantle earthquakes in the Himalayan collision zone. Geology, 47, 815–819. https://doi.org/10.1130/G46378.1
- Sheehan, A., de al Torre, T. L., Monsalve, G., Abers, G. A., & Hacker, B. R. (2014). Constraints from seismic attenuation and velocity tomography. Journal of Geophysical Research, 119, 567–580. https://doi.org/10.1002/2013JB010601
- Sippl, C., Schurr, B., Yuan, X., Schneider, F. M., Gadoev, M., Orunbaev, S., et al. (2013). Geometry of the Pamir-Hindu Kush intermediate-depth earthquake zone from local seismic data. *Journal of Geophysical Research*, 118, 1438–1457. https://doi.org/10.1002/ jgrb.50128
- Sloan, R. A., Jackson, J. A., McKenzie, D., & Priestley, K. (2011). Earthquake depth distribution in central Asia, and their relations with lithosphere thickness, shortening and extension. *Geophysical Journal International*, 185, 1–29. https://doi.org/10.1111/j.1365-246X.2010.04882.x
- Stearns, M. A., Hacker, B. R., Ratschbacker, L., Lee, J., Cottle, J. M., & Kylander-Clark, A. (2013). Synchronous Oligiocene-Miocene metamorphism of the Pamir and the north Himalaya driven by plate-scale dynamics. *Geology*, 41, 1071–1074. https://doi.org/10.1130/ G34451.1
- Subedi, S., Hetényi, G., Vergne, J., Bollinger, L., Lyon-Caen, H., Farra, V., et al. (2018). Imaging the Moho and the Main Himalayan Thrust in Western Nepal with receiver function. *Geophysical Research Letters*, 45, 13,222–13,230. https://doi.org/10.1029/2018GL080911
- Wang, C., Chen, W.-P., & Wang, L.-P. (2013). Temperature beneath Tibet. Earth and Planetary Science Letters, 375, 326–337. https://doi. org/10.1016/j.epsl.2013.05.052
- Webb, A. A. G., Guo, H., Clift, P. D., Husson, L., Müller, T., Constantino, D., et al. (2017). The Himalaya in 3D: Slab dynamics controlled mountain building and monsoon intensification. *Lithosphere*, 9, 637–651. https://doi.org/10.1130/L636.1
- Whipp Jr., D. M., Ehlers, T. A., Blythe, A. E., Huntington, K. W., Hodges, K. V., & Burbank, D. W. (2007). Plio-Quaternary exhumation history of the central Nepalese Himalaya: 2. Thermokinematic and thermochronometer age prediction model. *Tectonics*, 26, TC3003. https://doi.org/10.1029/2006TC001991
- Wright, T. J., Elliott, J. R., Wang, H., & Ryder, I. (2013). Earthquake cycle deformation and the Moho: Implications for the rheology of continental lithosphere. *Tectonophysics*, 609, 504–523. https://doi.org/10.1016/j.tecto.2013.07.029



- Yakovlev, P. V., Saal, A., Clark, M. K., Hong, C., Niemi, N. A., & Mallick, S. (2019). The geochemistry of Tibetan lavas: Spatial and temporal relationships, tectonic links and geodynamic implications. *Earth and Planetary Science Letters*, 520, 115–126. https://doi.org/10.1016/j. epsl.2019.04.032
- Zhu, D.-C., Zhao, Z.-D., Niu, Y., Dilek, Y., Hou, Z.-Q., & Mo, X.-X. (2013). The origin and pre-Cenozoic evolution of the Tibetan Plateau. *Gondwana Research*, 23, 1429–1454. https://doi.org/10.1016/j.gr.2012.02.002